Author’s response to reviews

Title: Recurrence and mortality according to Estrogen Receptor status for breast cancer patients undergoing conservative surgery. Ipsilateral breast tumour recurrence dynamics provides clues for tumour biology within the residual breast.

Authors:

Romano Demicheli (demicheli@istitutotumori.mi.it)
Ilaria Ardoino (ilaria.ardoino@istitutotumori.mi.it)
Patrizia Boracchi (patrizia.boracchi@unimi.it)
Danila Coradini (danila.coradini@unimi.it)
Roberto Agresti (agrestiroberto@istitutotumori.mi.it)
Cristina Ferraris (ferrariscristina@istitutotumori.mi.it)
Massimiliano Gennaro (gennaromassimiliano@istitutotumori.mi.it)
William JM Hrushesky (williamhrushesky@gmail.com)
Elia Biganzoli (elia.biganzoli@istitutotumori.mi.it)

Version: 6 Date: 16 November 2010

Author’s response to reviews:

Requested formatting changes have been made.